Author's response to reviews

Title: The Heart Health Study - Increasing cardiovascular risk assessment in Family Practice for first degree relatives of patients with premature ischaemic heart disease: a randomised controlled trial

Authors:

Nigel P Stocks Prof (nigel.stocks@adelaide.edu.au)
Jessica L Broadbent Ms (jessica.broadbent@adelaide.edu.au)
Michelle F Lorimer Ms (michelle.lorimer@adelaide.edu.au)
Derek P Chew Prof (derek.chew@flinders.edu.au)
Philip Tideman Prof (philip.tideman@flinders.edu.au)
Gary Wittert Prof (gary.wittert@adelaide.edu.au)
Philip Ryan Prof (philip.ryan@adelaide.edu.au)

Version: 3 Date: 6 January 2015

Author's response to reviews:

In response to an editorial email on the 06/01/2015 I have added line and page numbers to the document, removed page breaks, added all the figures to the main document, removed attached figures, added a section on competing interests, moved the authors contributions section, added a section about abbreviations and added a section about supporting data.